Author's response to reviews

Title: Validation of the modified Fresno Test: assessing physical therapists' evidence based practice knowledge and skills

Authors:

Julie K Tilson (tilson@usc.edu)

Version: 9 Date: 20 May 2010

Author's response to reviews:

Dear editorial staff,
Minor typographical errors were corrected.
Sincerely,
Julie Tilson
tilson@usc.edu